Author's response to reviews

Title: Pharmacological targeting of valosin containing protein (VCP) selectively kills canine lymphoma cells by inducing DNA damage

Authors:

Marie-Eve Nadeau (marie-eve.nadeau@umontreal.ca)
Charlène Rico (chalene.m.rico@gmail.com)
Mayra Tsoi (mayra.tsoi@umontreal.ca)
Mélanie Vivancos (vivancos_melanie@hotmail.fr)
Sabin Filimon (sabin.filimon@umontreal.ca)
Marilène Paquet (marilene.paquet@umontreal.ca)
Derek Boerboom (derek.boerboom@umontreal.ca)

Version: 4
Date: 25 August 2014

Author's response to reviews:

Editor required changes be made. Line numbers as well as page number were added.